Author’s response to reviews

Title: A meta-analysis of genome-wide data from five European isolates reveals an association of COL22A1, SYT1, and GABRR2 with serum creatinine level

Authors:

Cristian Pattaro (cristian.pattaro@eurac.edu)
Alessandro De Grandi (alessandro.degrandi@eurac.edu)
Veronique Vitart (veronique@hgu.mrc.ac.uk)
Caroline Hayward (Caroline.Hayward@hgu.mrc.ac.uk)
Andre Franke (franke@gmx.info)
Yurii S Aulchenko (i.aulchenko@erasmusmc.nl)
Asa Johansson (asa.johansson@genpat.uu.se)
Sarah H Wild (sarah.wild@ed.ac.uk)
Scott A Melville (melville@broadinstitute.org)
Aaron Isaacs (a.isaacs@erasmusmc.nl)
Ozren Polasek (opolasek@yahoo.co.uk)
David Ellinghaus (d.ellinghaus@ikmb.uni-kiel.de)
Ivana Kolcic (ikolcic@snz.hr)
Ute Nöthlings (u.noethlings@iem.uni-kiel.de)
Lina Zgaga (lina.zgaga@zamir.net)
Tatjana Zemunik (t.zemunik@bsb.mefst.hr)
Carsten Gnewuch (Carsten.Gnewuch@klinik.uni-regensburg.de)
Stefan Schreiber (s.schreiber@mucosa.de)
Susan Campbell (s.s.campbell@hgu.mrc.ac.uk)
Nick Hastie (n.hastie@hgu.mrc.ac.uk)
Mladen Boban (mladen.boban@mefst.hr)
Thomas Meitinger (meitinger@helmholtz-muenchen.de)
Ben A Oostra (b.oostra@erasmusmc.nl)
Peter Riegler (rieglerbz@gmail.com)
Cosetta Minelli (cosetta.minelli@eurac.edu)
Alan F Wright (alan.wright@hgu.mrc.ac.uk)
Harry Campbell (Harry.Campbell@ed.ac.uk)
Cornelia M van Duijn (c.vanduijn@erasmusmc.nl)
Ulf Gyllensten (ulf.gyllensten@genpat.uu.se)
James F Wilson (jim.wilson@hgu.mrc.ac.uk)
Michael Krawczak (krawczak@medinfo.uni-kiel.de)
Igor Rudan (irudan@hotmail.com)
Peter P Pramstaller (peter.pramstaller@eurac.edu)
the EUROSPAN consortium (no_email@here.com)

Version: 7 Date: 8 March 2010

Author’s response to reviews: see over
Dear Editorial Production Team,

Concerning our article “A meta-analysis of genome-wide data from five European isolates reveals an association of COL22A1, SYT1, and GABRR2 with serum creatinine level” (MS: 2008964933085808), we have added “the EUROSPAN consortium” as last author in the “author details” section in the electronic submission system. Unfortunately, the system did not allow us to leave the email box empty, so we provided a fake email address (no_email@here.com) that you should remove in the publication phase.

We hope that, with the provided corrections, the article is now suitable for publication. For any questions, please do not hesitate to contact me at cristian.pattaro@eurac.edu or +39 0471 055 527.

Yours sincerely,
Cristian Pattaro, on behalf of all the Authors.

Institute of Genetic Medicine
European Academy
39100 Bolzano-Bozen, Italy.
Tel. +39 0471 055 527
Fax +39 0471 055 599